Dear editor,

Thank you for your careful guidance of this article. Revision has been made according to the suggestions of the reviewer:

**Reviewer:** 06110819

Negative emotion can affect the hemodynamic fluctuation of patients undergoing interventional surgery, which is not conducive to the smooth and safe operation of interventional surgery. Programmed nursing formulates a series of nursing programs according to the characteristics of the interventional operating room. In addition, it intends to ensure the standardization, organization and rationality of nursing measures and to avoid the blindness and arbitrariness of nursing intervention. In this study, the authors explored the intervention effect of motivational psychological nursing combined with programmed nursing on compliance and bad mood of patients in interventional operating room. The study is very well designed, and the results are interesting. A minor editing is required. Some minor language polishing should be corrected.

**Answer:** The manuscript was edited carefully, and the language has been edited by an editing company again.

**Reviewer’s code:** 06074564

This is an interesting study of encouraging mental nursing and programmed nursing of interventional operating room patients. The manuscript is very well written. Comments: (1) A short background should be added to the abstract, and the results in the abstract should be shorten. (2) A conclusion should be added to the main text.
**Answer:** Thank you for your advice. (1) The background has been added to the abstract; and (2) conclusion has been added to the main text.

Thank you again for publishing our manuscript in the World Journal of Clinical Cases.

Sincerely Yours,

Huiping Mao